Author's response to reviews

Title: Web 2.0 systems supporting childhood chronic disease management: A pattern language representation of a general architecture

Authors:

Toomas Timpka (tti@ida.liu.se)
Henrik Eriksson (her@ida.liu.se)
Johnny Ludvigsson (johnny.ludvigsson@lio.se)
Joakim Ekberg (joakim.ekberg@ihs.liu.se)
Sam Nordfeldt (sam.nordfeldt@lio.se)
Lena Hanberger (lena.hanberger@lio.se)

Version: 6 Date: 27 November 2008

Author's response to reviews:

Dear Editor,

thank you for the final proofs! We have read them carefully without finding any significant errors or typos. However, for consistency, we will upload a slightly revised figure where 'Text box' has been changed to 'Appendix'. The manuscript is thereafter from our perspective ready for production.

Linköping, November 2008,

Yours,

Sincerely,

Toomas Timpka
Professor Toomas Timpka MD. PhD.
Section of Social Medicine and Public Health
Department of Medicine and Health
Faculty of Health Sciences
Linköping University
S-581 83 Linköping
Sweden
Telephone : +46 13 222383
Telefax : +46 13 221865
e-mail : tooti@ida.liu.se